Author's response to reviews

Title: The perception of Medical Students and academic staff about the characteristics of a good clinical teacher: an analytical cross-sectional study

Authors:
Ali Akbar A.A. Haghdoost (ahaghdoost@kmu.ac.ir)
Mohamad Reza M.R. Shakibi (mrshakibi@gmail.com)

Version: 2 Date: 10 October 2005

Author's response to reviews:

Dear Editor

I attached a manuscript on the perceptions of academic staff and students about the characteristics of positive role models in clinical settings in Iran. We believe this topic might be important because it helps clinical teachers to understand more deeply about the relationship between their characteristics and their performance. In addition, we believe that the dominant culture in a society plays an important role on students' perceptions about positive models and their characteristics. However, most of the publications on this topic were from western countries. Therefore, we assume that our paper may add valuable information about the perception of Iranian students and staff. From this point of view, findings of this research are by some means new.

According to the above explanation, I hope you would kindly consider our manuscript to be evaluated by your team and reviewers, and I wish our manuscript could be accepted to be published in your valuable journal.

Sincerely yours

Dr A.A. Haghdoost